Author's response to reviews

Title: Propensity Score Matching in Estimating the Effect of Managerial Education on Academic Planning Behavior

Authors:

Huy V Nguyen (nvanhuy@yahoo.com)
Minh An T Dao (daothiminhan@yahoo.com)
Dzung V Do (dovandzung@gmail.com)

Version: 2 Date: 30 October 2011

Author's response to reviews: see over
31-October-2011

Dear Editors,

Thank you very much for your comments. We have modified the manuscript in response to the suggestions.

1. Added the conclusion section after discussion.
2. The section on causal inferences was moved to the section after the theoretical framework is presented. The paragraph beginning ‘To review a higher education program’ was also moved into the section after the theoretical framework.
3. The section of methods was further shortened a bit so it would be more succinct.
4. Some information about authors was added in the manuscript.

We look forward to hearing from you soon.

Sincerely,

The First, Correspondent Author

Nguyen Van Huy
Institute for Preventive Medicine and Public Health, Hanoi Medical University, Vietnam
Tel: 844.38523798 (ext. 108) or Cell: +84.917363919
Email: nvanhuy@yahoo.com
Mailing address: Institute for Preventive Medicine and Public Health, Hanoi Medical University, 01 Ton That Tung Str, Dong Da Dist, Hanoi, Vietnam.